Author's response to reviews

Title: Validation of the Cognitive Assessment of Later Life Status (CALLS) instrument: a computerized telephonic measure

Authors:

Valerie C Crooks (valerie.c.crooks@kp.org)
Thomas D Parsons . (tparsons@ict.usc.edu)
J. Galen Buckwalter (galenbuckwalter@eharmony.com)

Version: 3 Date: 18 May 2007

Author's response to reviews: see over
May 18, 2007

Dr. Jo Appleford  
Senior Assistant Editor  
BMC – series journals - Neurology

RE: Validation of the Cognitive Assessment of Later Life Status (CALLS) instrument: a computerized telephonic measure  
MS 2051008537112725

Dear Dr. Appleford:

We are pleased that you are publishing our article. Regarding the one Discretionary Revisions, we elected to stay with our original introduction. We have altered the title according to your specifications and eliminated all underlines in section headings of the paper. We have also conducted a final edit of the paper and believe it is ready for press. Thank you for giving us the opportunity to do a final edit of the paper.

Sincerely,

Valerie C. Crooks